Author's response to reviews

Title: Five-year teaching evaluation of oral microbiology experiment curriculum

Authors:

Min Nie (niemin@whu.edu.cn)
yan Zhen Gao (973066427@qq.com)
yu Xin Wu (645933142@qq.com)
xi Chen Jiang (354454133@qq.com)
hui Jia Du (137711727@qq.com)

Version: 2
Date: 17 October 2014

Author's response to reviews:

We revise our manuscript as follows:
1. include line and page numbers.
2. include the email addresses of all authors in title page.